Author's response to reviews

Title: Mammoth interatrial septal aneurysm in the ICE age

Authors:

Ravinay Bhindi (ravinay.bhindi@cardiov.ox.ac.uk)
John Timperley (john.timperley@orh.nhs.uk)
Oliver J Ormerod (Oliver.ormerod@orh.nhs.uk)

Version: 6 Date: 22 September 2007

Author's response to reviews:

Dear Dr Picano

Please find attached our revised manuscript which now incorporates the 2 suggested changes:

1. ‘title’ is now removed from before the title
2. The abstract is included in the manuscript as it appears in the online submission system.

Regards

Dr Ravinay Bhindi